Author's response to reviews

Title: Assessment of Nutritional Status and Diet Adequacy in Senior Residents of Newfoundland and Labrador, Canada: a cross-sectional

Authors:

Jing Yan (yanjing@tijmu.edu.cn)
Lin Liu (ll3332@mun.ca)
Barbara Roebothan (broeboth@mun.ca)
Ann Ryan (annr@mun.ca)
Zhi Chen (zc1476@mun.ca)
Yanqing Yi (yanqing.yi@med.mun.ca)
Peizhong Wang (pwang@mun.ca)

Version: 2 Date: 27 September 2013

Author's response to reviews:

Ethical Consideration was added to the Methods section. And the word count was changed to 3231.